Author's response to reviews

Title: The Alberta Ambassador Program: delivering Health Technology Assessment results to rural practitioners

Authors:

Saifudin Rashiq (srashiq@ualberta.ca)
Pamela Barton (pamela.barton@calgaryhealthregion.ca)
Christa Harstall (christa.harstall@ahfmr.ab.ca)
Donald Schopflocher (donald.schopflocher@gov.ab.ca)
Paul Taenzer (paul.taenzer@calgaryhealthregion.ca)

Version: 4 Date: 15 March 2006

Author's response to reviews:

Title & Abstract - These are now identical in the manuscript and in the submission system.
Running title - Has been removed from the title page.
Underlining - Has been removed from section headings.
Keywords section - Has been removed.
Introduction - Has been renamed 'Background'.
References - References are now be cited in the text using consecutive normal case numbers in square bracket
References - The reference has been reformatted in the journal's style
Tables - Tables 1, 3 and 4 have been reformatted in accordance with your instructions